Author’s response to reviews

**Title:** Associations between skin barrier characteristics, skin conditions and health of aged nursing home residents: a multi-center prevalence and correlational study

**Authors:**

Elisabeth Hahnel (elisabeth.hahnel@charite.de)

Ulrike Blume-Peytavi (ulrike.blume-peytavie@charite.de)

Carina Trojahn (carina_trojahn@gmx.de)

Jan Kottner (jan.kottner@charite.de)

**Version:** 3 **Date:** 25 Oct 2017

**Author’s response to reviews:**

Thank you very much for giving us the opportunity for publication in BMC Geriatrics. All requested changes have been made accordingly.

**Editor Comments**

1. Please provide figure legends under a separate heading of 'Figure Legends' after the References. If Figure titles/legends are within the main text of the manuscript, please move them. Response: The requested changes were made accordingly.

2. Please remove the section “protocol” from your declarations section. Response: The requested changes were made accordingly.

3. Please remove your “registration” section from your declarations, and instead include this information in your “ethical approval and consent to participate” section. Response: The requested changes were made accordingly.

4. Please include your trial registration date at the end of your Abstract. Response: The requested changes were made accordingly.